Author's response to reviews

Title: Accuracy evaluation of CAD/CAM generated splints in orthognathic surgery: A cadaveric study.

Authors:

Thomas MD Schouman (thomas.schouman@psl.aphp.fr)
Philippe PhD Rouch (Philippe.ROUCH@ensam.eu)
Benoît MD,DMD Imholz (benoit.imholz@hcuge.ch)
Jean MD Fasel (Jean.Fasel@unige.ch)
Delphine PhD Courvoisier (delphine.courvoisier@hcuge.ch)
Paolo MD,DMD Scolozzi (paolo.scolozzi@hcuge.ch)

Version: 3 Date: 24 March 2015

Author's response to reviews: see over
Dear Editor,

You will find attached the revised manuscript entitled “Accuracy evaluation of CAD/CAM generated splints in orthognathic surgery: A cadaveric study.” which I wish to submit for publication in the *Head and Face Medicine*.

This is an original paper, which has not been published in print or electronic format, and is not under consideration by another publication or electronic medium.

Thank you in advance for your editorial help.

With my best regards,

Yours sincerely,

Paolo Scolozzi, MD, DMD
Point-by-point description of the changes made:

1. As your research involves humans please include a statement of ethical approval in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate. Any experimental research on humans must be in compliance with the Helsinki Declaration.

As a cadaveric study, this project was granted exemption by our regional ethical review board. This has been added as requested.

2. Please re-upload your figure files in the system. To be able to number your figures accordingly, please note that multi-panel figures (those with parts a, b, c, d etc.) should be submitted as a single composite file that contains all parts of the figure.

This has been changed as requested.

3. Please include a competing interests section at then end of the manuscript, before the reference list. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

This has been added as requested.

4. Authors' contributions: Please include an Authors' contributions section before the Acknowledgements and Reference list.

This has been added as requested.